Author's response to reviews

Title: Online Pestkoppenstoppen: Systematic and theory-based development of a web-based tailored intervention for adolescent cyberbully victims to combat and prevent cyberbullying

Authors:

Niels C.L. Jacobs (niels.jacobs@ou.nl)
Trijntje Völlink (trijntje.vollink@ou.nl)
Francine Dehue (francien.dehue@ou.nl)
Lilian Lechner (lilian.lechner@ou.nl)

Version: 3 Date: 3 April 2014

Author's response to reviews: see over
Dear sir/madam,

Please find enclosed the resubmission of an adapted manuscript, proposed for publication in BMC Public Health titled:

**Online Pestkoppenstoppen: Systematic and theory-based development of a web-based tailored intervention for adolescent cyberbully victims to combat and prevent cyberbullying**

Niels C.L. Jacobs, MSc; Department of Psychology, Open University of the Netherlands, Heerlen, Netherlands.

Trijntje Völlink, PhD; Department of Psychology, Open University of the Netherlands, Heerlen, Netherlands.

Francine Dehue, PhD; Department of Psychology, Open University of the Netherlands, Heerlen, Netherlands.

Prof. Lilian Lechner, PhD; Department of Psychology, Open University of the Netherlands, Heerlen, Netherlands.

Word count: 10169 (including title page, abstract, main text, acknowledgements and references)

Number of tables: 1

Number of figures: 8

Number of additional files: 1

In response to the requests of the Editorial Office from BMC Public Health, we have made the following changes in our manuscript:

1. We will forward the documentation (original and translated) concerning the ethical and funding approval to BMCSeriesEditorial@biomedcentral.com on Monday the 7th of April. Our study has undergone peer-review by the funding body (Onderwijsbewijs).
2. We would like to confirm that this is an ongoing study. Currently, we are collecting data from the target group. In about 2 months, the first analysis will start.

3. Until now we have published one article that is related to this study. This article was published in Journal of Adolescence, and describes a Delphi study on the determinants of ineffective and improved coping of cyberbullying victims. The reference to this article is:


4. We have changed the Competing Interest paragraph into: The authors declare that they have no competing interests.

5. On page 23, we have added a power calculation.

6. We have added the date of our registration at the Nederlands Trial Register to the Trial Registration Number on page 3.

With kind regards, on behalf of all co-authors,

Niels Jacobs, M.Sc.
Open University, The Netherlands
Faculty of Psychology
PO Box 2960, 6401 DL Heerlen,
The Netherlands
niels.jacobs@ou.nl
Tel. +3145 - 576 28 74